March 1, 2014.

The Biomed Central Editorial Team

**Objects:** MS: 1665371269118404-pleomorphic adenoma presenting with conductive hearing loss in ear canal: a case report and literature review. Dr. Maruyama et al.

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your comments.

**The comments about the style of manuscript**

Before we proceed, please do not label Table 1 as Figure 4 as this may cause confusion. Kindly revise this accordingly.

We are so sorry that we have not uploaded Figure 4. Table 1 shows symptoms and Figure 4 shows age of the patients and sites of the tumor in the canal.

If you have any further questions, we would be happy to answer them. Again, thank you for your fruitful comments.

Sincerely,

Ayako Maruyama, MD.
Department of Otolaryngology, Tokyo Medical and Dental University
1-5-45, Yushima, Bunkyo-ku
Tokyo 113-8519, Japan
Tel +81-3-5803-5308
Fax +81-3-3813-2134
Email maruyamaoto@gmail.jp